

# **Geophysical Research Letters**<sup>\*</sup>

# **RESEARCH LETTER**

10.1029/2022GL098513

#### **Key Points:**

- Marsh deposition decreases delta slope, creating feedbacks that alter the spatial deposition of clastic material
- A small addition of marsh material, almost doubled the area near sea level, which has significant implications for coastal restoration
- The interaction of marsh and clastic deposition creates a delta hypsometry more akin to global deltas than experiments without marsh

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### **Correspondence to:**

K. M. Sanks, ksanks@tulane.edu

#### **Citation:**

Sanks, K. M., Zapp, S. M., Silvestre, J. R., Shaw, J. B., Dutt, R., & Straub, K. M. (2022). Marsh sedimentation controls delta top morphology, slope, and mass balance. *Geophysical Research Letters*, 49, e2022GL098513. https://doi. org/10.1029/2022GL098513

Received 2 MAR 2022 Accepted 27 MAY 2022

#### **Author Contributions:**

Conceptualization: K. M. Sanks, S. M. Zapp, J. R. Silvestre, J. B. Shaw, K. M. Straub

Data curation: K. M. Sanks, S. M. Zapp, J. R. Silvestre, J. B. Shaw, R. Dutt, K. M. Straub Formal analysis: K. M. Sanks Funding acquisition: J. B. Shaw, K. M. Straub

Investigation: K. M. Sanks, S. M. Zapp, J. R. Silvestre, J. B. Shaw, K. M. Straub Methodology: K. M. Sanks, S. M. Zapp, J. R. Silvestre, J. B. Shaw, K. M. Straub Resources: J. B. Shaw, K. M. Straub Software: K. M. Sanks Supervision: J. B. Shaw, K. M. Straub Validation: K. M. Sanks Visualization: K. M. Sanks Writing – original draft: K. M. Sanks

© 2022. American Geophysical Union. All Rights Reserved.

# Marsh Sedimentation Controls Delta Top Morphology, Slope, and Mass Balance

# K. M. Sanks<sup>1,2</sup> , S. M. Zapp<sup>1,3</sup>, J. R. Silvestre<sup>2</sup>, J. B. Shaw<sup>1</sup>, R. Dutt<sup>2</sup>, and K. M. Straub<sup>2</sup>

<sup>1</sup>Department of Geoscience, University of Arkansas, Fayetteville, AR, USA, <sup>2</sup>Department of Earth and Environmental Sciences, Tulane University, New Orleans, LA, USA, <sup>3</sup>Department of Oceanography and Coastal Sciences, Louisiana State University, Baton Rouge, LA, USA

**Abstract** Rising sea levels, subsidence, and decreased fluvial sediment load threaten river deltas and their wetlands. However, the feedbacks between fluvial and non-fluvial (marsh) deposition remain weakly constrained. We investigate how non-riverine, elevation-controlled deposition typified by marshes impacts sediment partitioning between a delta's topset, coastal zone, and foreset by comparing a delta experiment with proxy marsh accumulation to a control. Marsh accumulation alters fluvial sediment distribution by decreasing the slope in the marsh window by ~50%, creating a 78% larger marsh zone. Fluvial incursions into the marsh window trap 1.3 times more clastic volume. The volume exported to deep water remains unchanged. Marsh deposition shifts elevation distributions toward sea level, which produces a hypsometry akin to field-scale deltas. The elevation-lowering effect of marshes on an equilibrium delta shown here constitutes an unexplored feedback and an important aspect of coastal sustainability.

**Plain Language Summary** Low-lying coastal zones, often with abundant vegetation (wetlands), are threatened worldwide because of rising sea level and decreased sediment supply. Coastal sediment accumulation is a fundamental process that helps these regions keep pace with rising sea level. This sediment may be delivered directly from rivers, or as mud and plant material in wetlands (e.g., marshes and mangrove forests) and shallow bays. Our study shows that sediment accumulated in the second manner alters the elevation distribution of coastal regions and the spatial deposition of the river sediment. These results provide important information for plans to help sustain and restore coastal land area.

# 1. Introduction

River deltas and their marsh platforms host diverse ecosystems threatened by anthropogenic impacts to coastal areas, such as rising sea levels, subsidence, and leveeing of channels (Ericson et al., 2006). Organic material production, a critical form of sediment accumulation in many river deltas, is the primary driver of marsh platform growth (Nyman et al., 2006), whereas clastic sedimentation via rivers drives deltaic lobe growth (Edmonds et al., 2009). To successfully predict the long-term fate of these ecosystems, the interaction controlling delta and marsh growth must be understood (Paola et al., 2011). While much is known about surface processes in channelized portions of river deltas (Edmonds & Slingerland, 2008; Li et al., 2017; Smart & Moruzzi, 1971) and much is known about sediment accumulation in marshes (Allen, 2000; Kirwan & Murray, 2007; Morris et al., 2002), the manner in which they interact remains largely uninvestigated.

Aerial imagery and the stratigraphic record show evidence of delta-marsh interaction in modern and ancient systems, and it is well known that deltaic channel deposits are sensitive to the deposition of fine-grained and organic material in floodplains (Bohacs & Suter, 1997; Esposito et al., 2017; Hoyal & Sheets, 2009). For example, ~25% of the recent Mississippi River Delta (MRD) sedimentation was organic marsh material (by mass) (Holmquist et al., 2018; Sanks et al., 2020). Further, evidence preserved in strata suggests organic-rich deposition influenced deltaic processes over most of the Phanerozoic (Chesnut & Greb, 1992). Both modern and ancient records suggest that clastic inputs influence the stability and growth of the marsh platform, thus influencing coastal sustainability.

The influence of marsh sediment accumulation on long-term (> $10^3$  years) delta dynamics remains largely unknown. A fundamental aspect of this sedimentation is that it is not directly sourced from the river and accumulates in deltas globally. This mass is composed of both mineral and organic material, has a low bulk density, and supplements clastic river sedimentation that is focused near channel mouths and within the channel network



(Sanks et al., 2020). The organic component is formed in-situ via primary production of plants and accumulates as a parabolic function of elevation relative to sea level, with maximum production occurring around mean high tide (Morris et al., 2002). As it stands, experimental deltas have not included a process for accumulating this sediment. Instead, sediment cohesion is sometimes used as a proxy for vegetation (Hoyal & Sheets, 2009; Li et al., 2017), but not always (Ganti et al., 2016). Vegetation hydrodynamics have also been studied in an experimental delta through the use of alfalfa (Piliouras et al., 2017), though the deltas formed in this study were too steep to be accurate representations of coastal deltas.

Here, we investigate the influence of non-riverine accumulation on delta morphology and mass balance by comparing two physical experiments conducted at the Tulane University Sediment Dynamics Laboratory. We incorporate proxy-marsh sediment accumulation in an experimental river delta, an important advance in experimental sedimentology and coastal restoration. We compare this experiment to a previous, identical experiment that formed without marsh sedimentation. This setup is ideal to understand the interaction of ecogeomorphic processes in coastal wetlands and physical processes of river deltas due to the ability to assess long-term behavior at reduced time and length scales, control on forcing conditions (Table 1), precise measurements, and autogenic dynamics (Paola et al., 2009). By analyzing the experiments over long timescales relative to autogenic dynamics, we can interpret any differences as direct results of marsh deposition.

# 2. Materials and Methods

# 2.1. Experimental Setup and Data

We investigate two experimental deltas formed under identical boundary conditions. The only difference is that the control experiment evolved without explicit marsh sedimentation, while the treatment experiment evolved with a marsh proxy (Table 1). Thus, any changes between the two experiments can be attributed directly to the addition of marsh.

Both experiments were run for 560 hr (~10 times the compensation timescale), which captures many channel avulsions and inherent stochasticity of the system (Wang et al., 2011). Thus, time series explored in this work can be considered ergodic and emergent dynamics within these time series are robust. LiDAR scans of the basin were collected every one (control) or two (treatment) hours while the experiments were paused. Aerial imagery was taken every 15 min.

The deposit was sectioned from distal to proximal along strike every 10 cm. We use image processing to obtain a stratigraphic marsh fraction roughly every 10 cm in strike (Figure S1 in Supporting Information S1), which was interpolated across the basin using Bayesian kriging techniques to estimate the marsh and clastic volume sequestered in the basin (Text S2 in Supporting Information S1).

# 2.2. Marsh Proxy

The marsh proxy introduces non-riverine deposition to the treatment experiment (Figure 1a). In total, the mass of marsh proxy added to the system is 8% of the riverine sediment input and makes up 15% of the final deposit volume. While we discuss this proxy in terms of organic sedimentation, it may also represent fine-grained deposition from non-riverine processes (e.g., tides, waves, and storms) in tidal flats and wetland platforms. For simplicity, the marsh proxy simulated only the sediment properties of organic material, neglecting some physical properties of vegetation (e.g., stem density). We use kaolinite (clay) as the marsh proxy, which has a low initial bulk density (~90% porosity when deposited in water), uniform deposition upon settling, and relatively high settling velocity when surfactant (0.01% Jet Dry to water input) is added. Further, a distinctly different grain size and color from the riverine sediment makes it ideal to analyze in aerial imagery (Figure 1a) and stratigraphy (Figure S1 in Supporting Information S1).

To first order, marshes accumulate as a function of elevation relative to sea level (rsl) (Baustian et al., 2012; Cahoon et al., 1995; Kirwan et al., 2010; Morris et al., 2002). This generalization simplifies many complex processes of marsh ecology (Morris et al., 2002) and trapping of fine sediment (Li et al., 2009), yet the vast swaths of coastal marsh within decimeters of sea level show that this is a dominant, emergent control. We adapt the marsh production model from Morris et al. (2002), which shows an optimum accumulation rate near rsl and suboptimal accumulation above and below. Experimental elevations were scaled to the emergent channel depth

# Table 1

The Experimental Conditions for Both the Control (No Marsh) and Treatment (Marsh) Experiments Used for Comparison in This Study

Boundary condition	Control	Treatment
Sediment Mixture	Hoyal and Sheets (2009)	Hoyal and Sheets (2009)
Realtive Sea Level Rise (RSLR <sub>b</sub> )	0.25 mm/hr	0.25 mm/hr
Riverine Sediment Discharge (Q <sub>s</sub> )	1.41 kg/hr	1.41 kg/hr
Riverine Water Discharge $(Q_w)$	1.72*10 <sup>-4</sup> m <sup>3</sup> /s	1.72*10 <sup>-4</sup> m <sup>3</sup> /s
In-situ Marsh Deposition $(Q_m)$	None	200 g/2-hr (average); 3.7 g/hex (max production); 1.7 g/hex (stable/unstable)

from the control experiment (~14 mm). Hence, generating three elevation zones that received marsh: -9 to -5 (unstable), -5 to 0 (maximum production), and 0 to 5 mm rsl (stable), and collectively represent the marsh window. The maximum production zone received enough kaolinite to accumulate ~1 times the base relative sea level rise rate (RSLR<sub>b</sub>; 0.5 mm/2-hr). The unstable and stable zones received enough sediment to accumulate ~0.5RSLR<sub>b</sub> (Figure 1b).

To apply the marsh proxy, LiDAR scans taken while the experiment was paused provide the median elevation of hexagonal grid cells covering the basin (146.14 cm<sup>2</sup>; 7.5 cm sides; Figure 1c). Depending on the median elevation of the cell, we deposit either 3.4 (maximum production zone) or 1.7 g (stable and unstable zones) of



Figure 1. (a) The silver cart (top of image) holds the marsh dispenser, which deposits kaolinite at the center of each hexagonal grid (c) with an average elevation in the marsh window every 2 hr. The brown sediment is the kaolinite marsh proxy. (b) The model, adapted from Morris et al. (2002), used to determine the marsh zone. (c) The hexagonal grid imposed upon a LiDAR scan of the basin (hour 250). (d) Modeled versus actual marsh deposition (g) each deposition cycle during the experiment with the average ~200 g/2-hr shown as a green dashed line.





**Figure 2.** (a) Delta top  $(\geq -9 \text{ mm rsl})$  and marsh window (-9-5 mm rsl) area for the control and treatment through time. (b) Time-integrated mean probability density of elevations relative to sea level, with one standard deviation shown for both experiments. (c) Box plots showing the time distribution of fluvial (>5 mm rsl) and marsh window delta slopes. (d) Mean elevation (mm) as a function of radial distance from the entrance channel (mm) integrated over space and time, with one standard deviation shown for both experiments.

kaolinite. The marsh sediment dispenser (a sieve) is attached to a cart that moves about the basin. On average, the sieve is 0.60 m above the sediment surface, though this varies through time and space as sediment accumulates on the delta. Deposition is promoted by using a ButtKicker<sup>TM</sup> to vibrate the sieve (black box left of sieve in Figure 1a), triggering kaolinite to deposit evenly within the cell (Movie S1 in Supporting Information S1). While we deposit with great spatial precision, the proxy mass is under-distributed by ~50% (Figure 1d; ~60 g/2-hr less than modeled). Thus, if we met the target rates, effects quantified here would have likely been accentuated. While less accurate than anticipated, the spatial deposition of sediment is consistent with the model from Morris et al. (2002) (Figure 1b). Thus the in-situ deposition of kaolinite provides a reasonable proxy for marsh accumulation, as shown by significantly altered morphology and clastic deposition between the experiments.

# 3. Results

### 3.1. Delta Morphology

A significant difference between treatment and control is observed in the area within the marsh window (-9 to 5 mm rsl) and the delta top ( $\geq$ -9 mm rsl). The marsh window was  $0.936 \pm 0.202$  m<sup>2</sup> in the control, but larger in the treatment at an average size of  $1.67 \pm 0.288$  m<sup>2</sup> (Figure 2a). Similarly, the delta top was smaller in the control at an average size of  $2.80 \pm 0.383$  compared to  $3.08 \pm 0.316$  m<sup>2</sup> for the treatment (Figure 2a). Considering the average delta top area, the treatment experiment was 10% larger than the control experiment, while the treatment marsh window was 78% larger.

# Table 2

The Clastic Volume Balance and Trapping Efficiency of Different Delta Regions for the Control and Treatment (Treat.) Experiments

Delta region	Area (m <sup>2</sup> ) [control]	Area (m <sup>2</sup> ) [treat.]	Clastic volume (m <sup>3</sup> ) [control]	Clastic volume (m <sup>3</sup> ) [treat.]	Trapping efficiency (%) [control]	Trapping efficiency (%) [treat.]
delta top ( $\geq -9 \text{ mm rsl}$ ; $\geq 50\%$ of time)	2.73	2.96	0.363	0.355	55.0	53.7
fluvial (>5 mm rsl; > 90% of time)	0.880	0.352	0.121	0.0413	18.3	6.25
marsh window (-9 to 5 mm rsl; $\geq 10\%$ of time)	2.87	4.28	0.339	0.453	51.4 [63.9 <sup>a</sup> ]	68.6 [73.2 <sup>a</sup> ]
off shore ( $<-9$ mm rsl; $\geq$ 50% of time)	N/A	N/A	0.297	0.306	100	100

Note. Treatment marsh sedimentation is excluded from volume.

<sup>a</sup>The trapping efficiency calculated using the volume of clastic sediment delivered to the marsh window instead of the clastic sediment delivered to the delta top. Refer to Text S2 in Supporting Information S1 for equations and further explanation.

The elevation distribution shows an increase in elevations within the marsh window in the treatment experiment (Figure 2b), suggesting a change in slope relative to the control. Slopes are calculated radially from the entrance channel by fitting a linear regression through the mean elevations of 5 mm wide radial transects. We observe a 20% larger slope in the fluvial region (>5 mm rsl) for the treatment (3.0%) as compared to the control (2.3%). Interestingly, the slope in the marsh window is significantly reduced from 3.7% in the control to 1.9% in the treatment experiment (Figure 2c).

The mean elevation as a function of radial distance from the entrance channel shows that the addition of the marsh proxy alters the elevation distribution of the delta top (Figure 2d). The treatment experiment has an increase in marsh window elevations and a decrease in the area of elevations in the fluvial region. The relative elevations in the fluvial region are also smaller (by about one channel depth on average). Further, the delta top slope decreases upon entrance to the marsh window in the treatment, which allows the marsh to persist over a greater distance.

# 3.2. Sediment Balance

To limit delta-marsh interaction, the fluvial region discussed below encompasses only the area above 5 mm relative to sea level (rsl) for at least 90% of the experiment. The marsh window is the area  $\leq 5$  and  $\geq -9$  mm rsl for greater than 10% of the experiment. Thus, the marsh window begins exactly where the fluvial zone ends. The delta top is the area that is  $\geq -9$  mm rsl for at least 50% of the experiment, which captures average conditions. Off-shore is  $\langle -9 \rangle$  mm rsl for at least 50% of the experiment.

While each experiment had the same clastic sediment input, the spatial distribution of sediment accumulation is different. For volume balance and trapping efficiency equations refer to Text S2 in Supporting Information S1. The area in the fluvial region for greater than 90% of the control experiment is  $0.880 \text{ m}^2$ , which accumulates  $0.121 \text{ m}^3$  of sediment throughout the experiment. The corresponding area of the treatment experiment is  $0.352 \text{ m}^2$ , which accumulates  $0.0413 \text{ m}^3$  of clastic sediment during the experiment. Since the marsh extent is larger in the treatment experiment, more clastic sediment is trapped in this elevation window than in the control (Figure S2 in Supporting Information S1). Thus, the marsh window has a 68.6% trapping efficiency (clastic sediment delivered to the delta top/clastic sediment accumulated in marsh) in the treatment, but a 51.4% trapping efficiency in the control (Table 2).

The area on the delta top is  $2.73 \text{ m}^2$ , accumulating a total volume of  $0.363 \text{ m}^3$  of sediment. The corresponding area of the treatment experiment is slightly larger ( $2.96 \text{ m}^2$ ), but accumulates less clastic sediment ( $0.355 \text{ m}^3$ ). Compared to the total fluvial input ( $0.660 \text{ m}^3$ ), this yields similar delta top trapping efficiencies of 55.0% in the control and 53.7% in the treatment (Table 2). Hence, similar amounts of clastic sediment are transported past the marsh zone. We also find that roughly 85% of the marsh deposited was preserved in the resulting delta top stratigraphy, which accounts for 15% of the delta top volume. Though the total clastic sediment sequestrated here is similar in both experiments, marsh sedimentation augments the clastic sedimentation in the treatment experiment leading to the formation of a vastly different delta.

# 4. Discussion

The experiments show that marshes interact with deltas and have first-order impacts on morphology and sediment partitioning. We show that even a small addition of marsh proxy sediment ( $\sim 8\%$  of riverine mass) drastically impacts delta formation. Specifically, marsh deposition flattens the delta, alters location of maximum clastic deposition, and changes the delta hypsometry.

# 4.1. An Important Feedback

It is remarkable that an 8% addition of marsh mass creates a 78% increase in extent of the marsh window. This marsh sedimentation is essential to the long-term stability of the treatment experiment. Paradoxically, the addition of marsh proxy reduces total clastic sedimentation on the delta top, but simultaneously bridges the gap to create a delta spanning a similar extent. This illustrates an important and previously unexplored feedback between marsh and river delta sediment accumulation.

The emergent effect of the interaction between marsh and clastic sedimentation is the decreased slope of the subaerial marsh window. Because the treatment experiment has smaller slopes from the shoreline to the top of the marsh window and the shoreline location changes only slightly (Figure 2d), the area from the top of the marsh window to the apex must be smaller in the treatment experiment (Figures 2c and 2d). Marshes do not erode sediment from upstream to include within the marsh window, yet the lower slopes of a marsh in dynamic equilibrium with its delta effectively "steal" clastic sediment from higher elevations. For example, the fluvial area accumulated 3 times less clastic volume in the treatment experiment (Table 2). Instead, the remaining sediment trapped on the delta top is sequestered in the marsh window, which accumulates 1.3 times more clastic volume (Table 2) than the control. While marsh deposition changes the sediment balance between the marsh window and elevations above it, the clastic sediment partitioning of the topset and foreset remains similar. Even so, the decreased slope and associated feedbacks leads to variation in spatial clastic deposition in the treatment experiment as compared to the control.

Decreased delta top slopes have previously been shown to alter delta morphology and increase channelization (Parker et al., 1998). Decreased delta slopes are a function of grain size and cohesion (Edmonds & Slingerland, 2010; Caldwell & Edmonds, 2014; Li et al., 2017), as well as a function of the ratio of water to sediment discharge (Powell et al., 2012; Whipple et al., 1998; Wickert et al., 2013). Here we suggest a new mechanism for lowering delta top slope: non-riverine sedimentation in the floodplain. The slope break caused by marshes has been shown to influence avulsion locations (Ratliff et al., 2021). Hence, this process matters for modern-day and ancient river deltas, which often support large swaths of marsh.

# 4.2. Delta Hypsometry

Equilibrium hypsometry, or the elevation distribution on the delta top, shows enhanced areas of elevations near sea level where marsh sedimentation or similar processes are present (Figure 2b). Using the ETOPO Global Relief Model (NOAA) in Google Earth Engine, we explore this hypothesis for four large river deltas (MRD, Ganges Brahmaputra Meghna Delta (GBMD), Mekong River Delta, and Rio Grande River Delta). Despite coarse resolution and systematic errors in this digital elevation model (Minderhoud et al., 2019), comparison at the vertical scale of several meters is appropriate (Text S3, Figure S3 in Supporting Information S1). Scaling by channel depth (for comparison across scales) reveals the general hypsometry of these deltas (Figure 3).

The treatment experiment and global deltas show a peak in elevations between 0 and 0.5 channel depths above relative sea level (rsl). In both the treatment experiment and global deltas, >30% of all elevations between -1 and 3 channel depths lie between 0 and 0.5 channel depths rsl, while the control experiment only has 15% of elevations here. Rather, the control experiment shows its peak around 0.8 channel depths rsl due to increased slopes and associated reduced area near the shoreline. The marsh proxy organizes the treatment experiment's hypsometry to reflect the dominant hypsometric feature of delta systems. At a minimum, this suggests that proxies for non-riverine, elevation-based coastal accumulation can improve the fidelity of laboratory scale models. It also suggests that purely fluvial, lobe based delta deposition is insufficient to understand sedimentation in modern deltas. While organic deposition is a reasonable control on these systems, tidal flat and barrier island reworking should also fundamentally influence delta hypsometry and sediment partitioning in similar ways, because they





**Figure 3.** Probability density function showing hypsometry of control and treatment experiment and four global deltas (Mississippi River Delta [MRD], Ganges Brahmaputra Meghna [GBMD], Mekong, and Rio Grande). Elevations are normalized by channel depth ( $\eta^*$ ) for comparison across scales.

are also focused deposition near sea level. The point here is that despite many different accumulation processes and ecologies in these field-scale deltas, the proxy is successful in pushing the experimental hypsometry closer to field systems. Succinctly, the coupling of riverine and non-riverine sediment deposition appears to be essential in shaping global deltas.

# 4.3. Limitations

The kaolinite marsh proxy is not yet capable of discerning the effect of distinct ecologies or non-fluvial processes, meaning the proxy reflects any elevation-based, non-riverine coastal deposition. Our study differs fundamentally from those that consider the hydrodynamic or sediment transport effects of vegetation (e.g., stem density and roughness). Instead, it is the proxy's volume that alters delta slopes and zones of mass accumulation. Though the influence of vegetation properties on hydrodynamics has been shown to impact clastic sedimentation rates (Nardin & Edmonds, 2014), this interaction occurs on much shorter timescales (daily tidal cycle) than the timescales focused on here (>10<sup>3</sup>). Similarly, the proxy is not able to capture erosional effects of wind and waves like marsh edge erosion (McLoughlin et al., 2015) or pond formation and expansion (Himmelstein et al., 2021). However, these processes also occur on shorter timescales than focused on here. If ecological, hydrodynamic, or erosional effects pushed sedimentation away from the elevation-based deposition model, then results may differ in ways that cannot be quantified here.

Our study is limited to the influence of non-riverine deposition on delta dynamics in an equilibrium system. We do not incorporate an acceleration of sea level rise rate (SLRR) thus, we cannot assess the limits of marsh to vertically aggrade or move laterally with changes in SLRR. The marsh proxy would likely interact dynamically with accelerating SLRR, and this can be assessed with future studies.

# 4.4. Implications

This work can be used to inform restoration and management plans on river deltas with significant marsh deposition. Successful restoration of deltaic wetlands hinges on understanding delta hypsometry and the temporal and spatial clastic sediment deposition rates. While marsh sedimentation is relatively continuous (once every 2 hr) in the marsh window, it is important to note that this region accumulates primarily fluvial sediment. The extent of this region is increased due to the feedbacks between the river and marsh. Given the importance of channel-marsh interaction to the mass balance in the treatment experiment and in the absence of other clastic sediment distribution mechanisms (e.g., tides or storms), limiting channel-marsh interaction via leveeing could significantly alter the feedbacks observed here.

Engineered marsh platforms must be consistent with how the wetland platform would grow naturally (Paola et al., 2011). Modern deltas have elevation windows that matter for habitability (higher elevation, fluvial ridges) and others that matter for storm surge protection and biodiversity (lower elevation, wetlands). The presence of non-riverine deposition on the shallow platforms created via river diversions (or other restoration methods) will create mostly land at or near sea level. Millions of people globally rely on this region to mitigate flooding (Edmonds et al., 2020), thus the probability distribution of elevations (Figure 2b) will eventually have implications for the extent of storm surges and susceptibility to drowning. Similarly, the change in coastal accumulation rates seen in the treatment experiment has implications for the abiotic, fluvial deposit (i.e., fluvial ridges). Since the interaction between rivers and wetlands controls this area partitioning, it should be a significant control on modern deltas and any future river diversions created to support them.

# 5. Conclusion

We show that the addition of marsh proxy sedimentation in a delta experiment fundamentally alters the mass balance and hypsometry of the resulting delta. Specifically, we find a new control on delta top slope: non-riverine sedimentation. The decreased marsh window slope creates feedbacks that impact the spatial and temporal distribution of riverine sediment, leading to increased area near sea level. The interaction of river and marsh sediment in the treatment experiment leads to a morphological signature more consistent with modern-day river deltas than the control. Since coastal wetlands accumulate sediment to keep pace with relative sea level rise in the low-lying regions of deltas, they fundamentally flatten land near the coast creating the vast platforms seen globally. The lower slopes create feedbacks with clastic sediment deposition patterns that will help to inform future restoration plans, as these plans typically hinge on the successful distribution and retention of riverine sediment.

# **Data Availability Statement**

Data used to reproduce the results of this study are archived at: https://figshare.com/articles/dataset/TDWB\_19\_2\_ MassBalance\_Morphology/19963034 and Sanks (2022a). These data must be downloaded and placed in the data folder of the GitHub repository (https://github.com/kmsanks/TDWB\_19\_2\_MassBalance\_Morphology). The repository contains the software used to reproduce the results of this study and is archived on both GitHub and in Zenodo (Sanks, 2022b). Data archiving of the raw experimental data is available for both TDB-18-1 (control; Straub & Dutt, 2022) and TDWB-19-2 (treatment; Sanks et al., 2022). Note, this data is not needed to reproduce any results from the study, but may be of interest for other researchers.

# References

Allen, J. (2000). Morphodynamics of holocene salt marshes: A review sketch from the Atlantic and Southern North Sea coasts of Europe. *Quaternary Science Reviews*, 19(12), 1155–1231. https://doi.org/10.1016/S0277-3791(99)00034-7

Baustian, J. J., Mendelssohn, I. A., & Hester, M. W. (2012). Vegetation's importance in regulating surface elevation in a coastal salt marsh facing elevated rates of sea level rise. *Global Change Biology*, 18(11), 3377–3382. https://doi.org/10.1111/j.1365-2486.2012.02792

Bohacs, K., & Suter, J. (1997). Sequence stratigraphic distribution of coaly rocks: Fundamental controls and paralic examples. *AAPG Bulletin*, 81(10), 1612–1639. https://doi.org/10.1306/3B05C3FC-172A-11D7-8645000102C1865D

Cahoon, D. R., Reed, D. J., & Day, J. W. (1995). Estimating shallow subsidence in microtidal salt marshes of the southeastern United States: Kaye and Barghoorn revisited. *Marine Geology*, 128(1–2), 1–9. https://doi.org/10.1016/0025-3227(95)00087-F

Caldwell, R. L., & Edmonds, D. A. (2014). The effects of sediment properties on deltaic processes and morphologies: A numerical modeling study. Journal of Geophysical Research: Earth Surface, 119, 961–982. https://doi.org/10.1002/2013JF002965

Chesnut, J., & Greb, S. F. (1992). Lowstand versus highstand eustatic models for peat preservation: The coal-bearing rocks of the Breathitt Group, Eastern Kentucky. *Geological Society of America, Abstracts with Programs (United States)*, 24, 7.

Edmonds, D. A., Caldwell, R. L., Brondizio, E. S., & Siani, S. M. O. (2020). Coastal flooding will disproportionately impact people on river deltas. *Nature Communications*, 11(1), 4741. https://doi.org/10.1038/s41467-020-18531-4

Edmonds, D. A., Hoyal, D. C. J. D., Sheets, B. A., & Slingerland, R. L. (2009). Predicting delta avulsions: Implications for coastal wetland restoration. *Geology*, 37(8), 759–762. https://doi.org/10.1130/G25743A.1

Edmonds, D. A., & Slingerland, R. L. (2008). Stability of delta distributary networks and their bifurcations. *Water Resources Research*, 44(9), W09426. https://doi.org/10.1029/2008WR006992

#### Acknowledgments

The project was funded by an NSF grant (co PIs Kyle Straub; NSF EAR-1848994 that funded Kyle Straub and Jose Silvestre's time plus much of the experimental costs and John Shaw; NSF EAR-1848993 that funded John Shaw and Sam Zapp's time plus some of the experimental costs). We have no known conflicts of interest. We would like to thank Dr. Eric Barefoot for his monumental help in automating the treatment experiment. J. B. Shaw and K. M. Straub funded by the National Science Foundation.

- Edmonds, D. A., & Slingerland, R. L. (2010). Significant effect of sediment cohesion on delta morphology. *Nature Geoscience*, 3(2), 105–109. https://doi.org/10.1038/ngeo730
- Ericson, J., Vorosmarty, C., Dingman, S., Ward, L., & Meybeck, M. (2006). Effective sea-level rise and deltas: Causes of change and human dimension implications. *Global and Planetary Change*, 50(1–2), 63–82. https://doi.org/10.1016/j.gloplacha.2005.07.004
- Esposito, C. R., Shen, Z., Törnqvist, T. E., Marshak, J., & White, C. (2017). Efficient retention of mud drives land building on the Mississippi Delta plain. *Earth Surface Dynamics*, 5(3), 387–397. https://doi.org/10.5194/esurf-5-387-2017
- Ganti, V., Chadwick, A. J., Hassenruck Gudipati, H. J., & Lamb, M. P. (2016). Avulsion cycles and their stratigraphic signature on an experimental backwater-controlled delta. Journal of Geophysical Research: Earth Surface, 121, 1651–1675. https://doi.org/10.1002/2016JF003915
- Himmelstein, J., Vinent, O. D., Temmerman, S., & Kirwan, M. L. (2021). Mechanisms of pond expansion in a rapidly submerging marsh. Frontiers in Marine Science, 8, 1228. https://doi.org/10.3389/fmars.2021.704768
- Holmquist, J. R., Windham-Myers, L., Bliss, N., Crooks, S., Morris, J. T., Megonigal, J. P., et al. (2018). Accuracy and precision of tidal wetland soil carbon mapping in the conterminous United States. *Scientific Reports*, 8(1), 9478. https://doi.org/10.1038/s41598-018-26948-7
- Hoyal, D. C. J. D., & Sheets, B. A. (2009). Morphodynamic evolution of experimental cohesive deltas. Journal of Geophysical Research, 114, F02009. https://doi.org/10.1029/2007JF000882
- Kirwan, M. L., Guntenspergen, G. R., D'Alpaos, A., Morris, J. T., Mudd, S. M., & Temmerman, S. (2010). Limits on the adaptability of coastal marshes to rising sea level: Ecogeomorphic limits to wetland survival. *Geophysical Research Letters*, 37(23), L23401. https://doi. org/10.1029/2010GL045489
- Kirwan, M. L., & Murray, A. B. (2007). A coupled geomorphic and ecological model of tidal marsh evolution. Proceedings of the National Academy of Sciences, 104(15), 6118–6122. https://doi.org/10.1073/pnas.0700958104
- Li, Q., Benson, W. M., Harlan, M., Robichaux, P., Sha, X., Xu, K., & Straub, K. M. (2017). Influence of sediment cohesion on deltaic morphodynamics and stratigraphy over basin-filling time scales. *Journal of Geophysical Research: Earth Surface*, 122, 1808–1826. https://doi. org/10.1002/2017JF004216
- Li, S., Wang, G., Deng, W., Hu, Y., & Hu, W. (2009). Influence of hydrology process on wetland landscape pattern: A case study in the Yellow River Delta. *Ecological Engineering*, 35(12), 1719–1726. https://doi.org/10.1016/j.ecoleng.2009.07.009
- McLoughlin, S. M., Wiberg, P. L., Safak, I., & McGlathery, K. J. (2015). Rates and forcing of marsh edge erosion in a shallow coastal bay. *Estuaries and Coasts*, 38(2), 620–638. https://doi.org/10.1007/s12237-014-9841-2
- Minderhoud, P. S. J., Coumou, L., Erkens, G., Middelkoop, H., & Stouthamer, E. (2019). Mekong delta much lower than previously assumed in sea-level rise impact assessments. *Nature Communications*, 10(1), 3847. https://doi.org/10.1038/s41467-019-11602-1
- Morris, J. T., Sundareshwar, P. V., Nietch, C. T., Kjerfve, B., & Cahoon, D. R. (2002). Responses of coastal wetlands to rising sea level. *Ecology*, 83(10), 2869–2877. https://doi.org/10.1890/0012-9658
- Nardin, W., & Edmonds, D. A. (2014). Optimum vegetation height and density for inorganic sedimentation in deltaic marshes. *Nature Geoscience*, 7(10), 722–726. https://doi.org/10.1038/ngeo2233
- Nyman, J. A., Walters, R. J., Delaune, R. D., & Patrick, W. H. (2006). Marsh vertical accretion via vegetative growth. Estuarine. *Coastal and Shelf Science*, 69(3), 370–380. https://doi.org/10.1016/j.ecss.2006.05.041
- Paola, C., Straub, K., Mohrig, D., & Reinhardt, L. (2009). The "unreasonable effectiveness" of stratigraphic and geomorphic experiments. *Earth-Science Reviews*, 97(1), 1–43. https://doi.org/10.1016/j.earscirev.2009.05.003
- Paola, C., Twilley, R. R., Edmonds, D. A., Kim, W., Mohrig, D., Parker, G., et al. (2011). Natural processes in delta restoration: Application to the Mississippi Delta. Annual Review of Marine Science, 3(1), 67–91. https://doi.org/10.1146/annurev-marine-120709-142856
- Parker, G., Paola, C., Whipple, K. X., & Mohrig, D. (1998). Alluvial fans formed by channelized fluvial and sheet flow. I: Theory. Journal of Hydraulic Engineering, 124(10), 985–995. https://doi.org/10.1061/(ASCE)0733-9429(1998)124:10(985)
- Piliouras, A., Kim, W., & Carlson, B. (2017). Balancing aggradation and progradation on a vegetated Delta: The importance of fluctuating discharge in depositional systems. Journal of Geophysical Research: Earth Surface, 122, 1882–1900. https://doi.org/10.1002/2017JF004378
- Powell, E. J., Kim, W., & Muto, T. (2012). Varying discharge controls on timescales of autogenic storage and release processes in fluvio-deltaic environments: Tank experiments. *Journal of Geophysical Research*, 117, F02011. https://doi.org/10.1029/2011JF002097
- Ratliff, K. M., Hutton, E. W. H., & Murray, A. B. (2021). Modeling long-term delta dynamics reveals persistent geometric river avulsion locations. *Earth and Planetary Science Letters*, 559, 116786. https://doi.org/10.1016/j.epsl.2021.116786
- Sanks, K. (2022a). TDWB\_19\_2\_MassBalance\_Morphology. https://doi.org/10.6084/m9.figshare.19963034
- Sanks, K. (2022b). kmsanks/TDWB\_19\_2\_MassBalance\_Morphology: v1.0.2 (v.1.0.2). Zenodo. https://doi.org/10.5281/zenodo.6607658
- Sanks, K., Zapp, S., Silvestre, J., Shaw, J., & Straub, K. (2022). TDWB-19-2-Surface-Processes. SEAD Internal Repository. http://doi. org/10.26009/s0UQYZ0M
- Sanks, K. M., Shaw, J. B., & Naithani, K. (2020). Field-based estimate of the sediment deficit in coastal Louisiana. Journal of Geophysical Research: Earth Surface, 125, e2019JF005389. https://doi.org/10.1029/2019JF005389

Smart, J. S., & Moruzzi, V. L. (1971). *Quantitative properties of delta channel networks* (Tech. Rep.). IBM Thomas J Watson Research Center. Straub, K., & Dutt, R. (2022). *TDB-18*. SEAD Internal Repository. http://doi.org/10.26009/s0G2SM3L

- Wang, Y., Straub, K. M., & Hajek, E. A. (2011). Scale-dependent compensational stacking: An estimate of autogenic time scales in channelized sedimentary deposits. *Geology*, 39(9), 811–814. https://doi.org/10.1130/G32068.1
- Whipple, K., Parker, G., Paola, C., & Mohrig, D. (1998). Channel dynamics, sediment transport, and the slope of alluvial fans: Experimental study. *The Journal of Geology*, 106(6), 677–694. https://doi.org/10.1086/516053
- Wickert, A. D., Martin, J. M., Tal, M., Kim, W., Sheets, B., & Paola, C. (2013). River channel lateral mobility: Metrics, time scales, and controls. Journal of Geophysical Research: Earth Surface, 118, 396–412. https://doi.org/10.1029/2012JF002386

# **References From the Supporting Information**

Diggle, P. J., & Ribeiro, P. J. (2001). geoR: A package for geostatistical analysis.

- Diggle, P. J., & Ribeiro, P. J. (2007). Bayesian inference. In Model-based geostatistics (pp. 157–198). Springer. https://doi. org/10.1007/978-0-387-48536-2
- Jerolmack, D. J., & Sadler, P. (2007). Transience and persistence in the depositional record of continental margins. *Journal of Geophysical Research*, 112(F3), F03S13. https://doi.org/10.1029/2006JF000555